Author’s response to reviews

Title: Health status of children left behind in rural areas of Sichuan Province of China: a cross-sectional study

Authors:

Daisheng Tang (dstang@bjtu.edu.cn)
Weng Choi (mb65838@connect.umac.mo)
Liyuan Deng (dlytrans@163.com)
Ying Bian (bianying@umac.mo)
Hao Hu (haohu@um.edu.mo; polo.haohu@outlook.com)

Version: 1 Date: 03 May 2018

Author’s response to reviews:

Dear editor:

Thanks for your email! We have made necessary modification according to your comments.

Best,